comp150 – Fall 2002 Handout: Exam3-Info Introduction to Computer Science Michael Goldwasser Loyola University Chicago Tuesday, 26 November 2002

Exam 3 Information

• The final exam for this course will be on Tuesday, 10 December 2002, from 10:20am–12:20pm.

• This exam is closed book and no calculating devices of any type will be allowed. You are allowed, however, to prepare in advance the back of this page with whatever notes you wish to place on it, and you may use this page during the exam. When the exam is over, staple this sheet to the rest of your exam.

• The format of the midterm will be as follows. There will be 14 questions available on the exam. Each student must choose to solve exactly 12 of the 14 questions. Students are free to read all of the questions, and make attempts, before deciding which 12 to select. However the student must explicitly mark which 12 problems are to be graded (if this is not clear, we will assume that the first 12 problems are to be graded).

• The potential coverage of material will be as follows. At least 50% of the questions will cover “new” material, that is topics which were introduced since the last exam. Specifically, it will included coverage of lectures from Tuesday, 29 Oct 2002 through Tuesday, 3 Dec 2002, as well as the associated readings and homeworks for those topics (please see the class schedule for exact coverage).

The remainder of the exam will be cumulative, revisiting major topics from the first two-thirds of the course. For the sake of simplicity, we will not included any questions involving Pep7 machine code or assembly language.